Dear BMC Public Health Editors,

according to the indication of Victorino Silvestre's email dated on March 27th, I am submitting a revision of the paper entitled "Chronic disease prevalence from Italian administrative databases in the VALORE project: a validation through comparison of population estimates with general practice databases and national survey".

In the revision the following changes have been performed:

1) added section 'Competing interests'  
2) added section 'Ethical approval'  
3) added section 'Author's contributions'  
4) changed the name of one of the authors: Mariadonata Bellentani instead of Maria Donata Bellentani

Best regards,